Author’s response to reviews

Title: Prevalence of cognitive impairment in individuals aged over 65 in an urban area. The importance of certain personal factors. The DERIVA study.

Authors:

Emiliano Rodriguez-Sanchez (emilianorodriguezsanchez@yahoo.es)
Sara Mora-Simón (saramorasimon@gmail.com)
María C Patino-Alonso (carpatino@usal.es)
Ricardo García-García (rigar@usal.es)
Alfonso Escribano-Hernández (aescribanoh@saludcastillayleon.es)
Luis García-Ortiz (Lgarciao@usal.es)
Mª Victoria Perea-Bartolomé (vperea@usal.es)
Manuel A Gómez-Marcos (magomez@usal.es)

Version: 5 Date: 15 November 2011

Author’s response to reviews: see over
Dear Editor:

Following the editorial suggestions, we enclose a new version of our manuscript entitled: “Prevalence of cognitive impairment in individuals aged over 65 in an urban area. The importance of certain personal factors. The DERIVA study”, Re: 9557193555848233, together with our replies to all the issues raised.

We have made the formatting changes required

The list of authors in the manuscript has been written exactly as they are in the submission system.

1. We have modified abstract structure
2. We have revised all references. The reference 36 is inserting in manuscript in page 9, line 8 and page 11 line 15.
3. We have removed qualifications of authors.
4. We have removed the visible vertical lines in tables.
5. We have change 'PARTICIPANTS AND METHODS' for Methods

Please contact us if you have any problems or questions.
Best wishes,

Emiliano Rodriguez